Fundamentals of Computer Programming 2

Practice Problems – January 6, 2003

Instructor: Ravi Kant Ryerson 177 e-mail: ravikant@cs.uchicago.edu Office Hours : Wed, Fri : 1:00PM - 1:45PM

TA: Xiaofei He Eck 2-B e-mail: xiaofei@cs.uchicago.edu

Important announcement : Maclab C++ tutorials are scheduled for the following days -

Thursday January 9 at 5:00 PM. Friday January 10 at 4:00 PM.

Here are some problems that require simple input/ouput and some basic arithmetic:

1. Write a program that asks the user how old s/he is and outputs the year in which s/he was born.

2. Write a program that takes as input the temperature in degrees Celsius and outputs the temperature in degrees Fahrenheit.

3. Write a program that takes as input the length and breadth of a rectangle and outputs its area.